Author’s response to reviews

Title: The Warwick Patellofemoral Arthroplasty Trial: a randomised clinical trial of total knee arthroplasty versus patellofemoral arthroplasty in patients with severe arthritis of the patellofemoral joint.

Authors:

Michelle Odumenya (michelle.odumenya@warwick.ac.uk)
Katie McGuinness (katie.mcguinness@warwick.ac.uk)
Juul Achten (j.achten@warwick.ac.uk)
Nick Parsons (nick.parsons@warwick.ac.uk)
Tim Spalding (tim@timspalding.com)
Matthew Costa (matthew.costa@warwick.ac.uk)

Version: 2 Date: 18 October 2011

Author’s response to reviews:

Dear Sir/Madam

The following revisions were made to the study manuscript:

1. Discussion section added to abstract
2. Titles of authors removed
3. 'Protocol for the PAT study' title removed
4. Figures removed from main manuscript
5. Headers of figures placed at end of manuscript after reference list
6. Typo removed from Randomisation paragraph

The rest of the manuscript remains in its original format. No information has been added or deleted.

Yours sincerely

Miss Michelle Odumenya